Author's response to reviews

Title: Infectious sacroiliitis: a retrospective, multicentre study of 39 adults.

Authors:

Marion Hermet (marionhermet@gmail.com)
Emeline Minichiello (minichielloemeline@gmail.com)
René Marc Flipo (rene-marc.flipo@chru-lille.fr)
Jean Jacques Dubost (jjdubost@chu-clermontferrand.fr)
Yannick Allanore (yannick.allanore@cch.app.fr)
Jean Marc Ziza (jmziza@hopitalcss.org)
Philippe Gaudin (pgaudin@chu-grenoble.fr)
Thierry Thomas (hierry.thomas@chu-st-etienne.fr)
Emmanuelle Demis (edernis@ch-lemans.fr)
Baptiste Glace (bglace@chu-clermontferrand.fr)
Alain Regnier (alainregnier@ch-vichy.fr)
Martin Soubrier (msoubrier@chu-clermontferrand.fr)

Version: 6 Date: 29 October 2012

Author's response to reviews: see over
Dear Dr Harris,

Please find enclosed our revised manuscript entitled “Infectious sacroiliitis: a retrospective, multicentre study of 39 patients and literature review”, which is an original article for publication in BMC infectious diseases.

We thank reviewers for their comments and we are sending you herewith our replies:

Comments:

- We’ve revised our ethics statement: “This study did not require formal ethical approval under French research regulations”
- We’ve obscured the date of birth seen on the figure 2 not to compromise patient confidentiality.

We hope that the editorial board and reviewers will consider this revised version suitable for publication in BMC infectious diseases.

Sincerely,

Marion Hermet, M.D. Martin Soubrier, Ph.D.

Corresponding author: Marion Hermet
marionhermet@gmail.com
Tel number: +0033678430495
Fax number: +0033473751436